Massachusetts Institute of Technology Handout 41 6.042J/18.062J: Mathematics for Computer Science November 2, 1995 Professors David Karger and Tom Leighton

Quiz Information

QUIZ 2

When: November 15, Wednesday, 7:00 to 9:00 PM Where: 10-250 Crib Sheet: The quiz is not open-book, but you are allowed to bring two

sheets to the exam. You may have whatever information you like on it but it must be handwritten and must not be xeroxed.

Syllabus: Everything done from the beginning until and including the tutorial

on November 13. But the emphasis will be on stuff done after Quiz 1.

Make-up Exam: If you have a conflict come see us NOW Special office hours: There will be special office hours during quiz-week. Your

TAs will give you details next week. We will also be handing out a practice quiz.